

Open access • Posted Content • DOI:10.1101/384909

Exploring Various Polygenic Risk Scores for Basal Cell Carcinoma, Cutaneous Squamous Cell Carcinoma and Melanoma in the Phenomes of the Michigan Genomics Initiative and the UK Biobank — Source link [2]

Lars G. Fritsche, Lauren J. Beesley, Peter VandeHaar, Robert B. Peng ...+14 more authors

Institutions: University of Michigan, University of Southern California

Published on: 04 Aug 2018 - bioRxiv (Cold Spring Harbor Laboratory)

Topics: Biobank and Population

Related papers:

- Exploring Various Polygenic Risk Scores for Skin Cancer in the Phenomes of the Michigan Genomics Initiative and the UK Biobank with a Visual Catalog:PRSWeb
- Association of Polygenic Risk Scores for Multiple Cancers in a Phenome-wide Study: Results from The Michigan Genomics Initiative.
- Cancer PRSweb: An Online Repository with Polygenic Risk Scores for Major Cancer Traits and Their Evaluation in Two Independent Biobanks.
- Cross-Cancer Evaluation of Polygenic Risk Scores for 17 Cancer Types in Two Large Cohorts
- PheWAS and Beyond: The Landscape of Associations with Medical Diagnoses and Clinical Measures across 38,662
 Individuals from Geisinger





GOPEN ACCESS

Citation: Fritsche LG, Beesley LJ, VandeHaar P, Peng RB, Salvatore M, Zawistowski M, et al. (2019) Exploring various polygenic risk scores for skin cancer in the phenomes of the Michigan genomics initiative and the UK Biobank with a visual catalog: *PRSWeb*. PLoS Genet 15(6): e1008202. <u>https://</u> doi.org/10.1371/journal.pgen.1008202

Editor: Xihong Lin, Harvard School of Public Health, UNITED STATES

Received: September 17, 2018

Accepted: May 17, 2019

Published: June 13, 2019

Copyright: © 2019 Fritsche et al. This is an open access article distributed under the terms of the <u>Creative Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: Data cannot be shared publicly due to patient confidentiality. The data underlying the results presented in the study are available from University of Michigan Medical School Central Biorepository at <u>https://research. medicine.umich.edu/our-units/centralbiorepository/get-access</u> and from the UK Biobank at <u>http://www.ukbiobank.ac.uk/register-apply/</u> for researchers who meet the criteria for access to confidential data. RESEARCH ARTICLE

Exploring various polygenic risk scores for skin cancer in the phenomes of the Michigan genomics initiative and the UK Biobank with a visual catalog: *PRSWeb*

Lars G. Fritsche^{1,2®}*, Lauren J. Beesley^{1®}, Peter VandeHaar^{1,2}, Robert B. Peng¹, Maxwell Salvatore¹, Matthew Zawistowski^{1,2}, Sarah A. Gagliano Taliun^{1,2}, Sayantan Das^{1,2}, Jonathon LeFaive^{1,2}, Erin O. Kaleba³, Thomas T. Klumpner^{3,4}, Stephanie E. Moser³, Victoria M. Blanc⁵, Chad M. Brummett^{3,4}, Sachin Kheterpal^{3,4}, Gonçalo R. Abecasis^{1,2}, Stephen B. Gruber⁶, Bhramar Mukherjee^{1,2,7,8,9}*

 Department of Biostatistics, University of Michigan School of Public Health, Ann Arbor, Michigan, United States of America, 2 Center for Statistical Genetics, University of Michigan School of Public Health, Ann Arbor, Michigan, United States of America, 3 Division of Pain Medicine, Department of Anesthesiology, University of Michigan Medical School, Ann Arbor, Michigan, United States of America, 4 Institute for Healthcare Policy and Innovation, University of Michigan, Ann Arbor, Michigan, United States of America, 5 Central Biorepository, University of Michigan Medical School, Ann Arbor, Michigan, United States of America, 6 USC Norris Comprehensive Cancer Center, University of Southern California, Los Angeles, California, United States of America, 7 Michigan Institute for Data Science, University of Michigan, Ann Arbor, Michigan, United States of America, 8 Department of Epidemiology, University of Michigan School of Public Health, Ann Arbor, Michigan, United States of America, 9 University of Michigan Rogel Cancer Center, University of Michigan, Ann Arbor, Michigan, United States of America,

• These authors contributed equally to this work.

* larsf@umich.edu (LGF); bhramar@umich.edu (BM)

Abstract

Polygenic risk scores (PRS) are designed to serve as single summary measures that are easy to construct, condensing information from a large number of genetic variants associated with a disease. They have been used for stratification and prediction of disease risk. The primary focus of this paper is to demonstrate how we can combine PRS and electronic health records data to better understand the shared and unique genetic architecture and etiology of disease subtypes that may be both related and heterogeneous. PRS construction strategies often depend on the purpose of the study, the available data/summary estimates, and the underlying genetic architecture of a disease. We consider several choices for constructing a PRS using data obtained from various publicly-available sources including the UK Biobank and evaluate their abilities to predict not just the primary phenotype but also secondary phenotypes derived from electronic health records (EHR). This study was conducted using data from 30,702 unrelated, genotyped patients of recent European descent from the Michigan Genomics Initiative (MGI), a longitudinal biorepository effort within Michigan Medicine. We examine the three most common skin cancer subtypes in the USA: basal cell carcinoma, cutaneous squamous cell carcinoma, and melanoma. Using these PRS for various skin cancer subtypes, we conduct a phenome-wide association study (PheWAS) within the MGI data to evaluate PRS associations with secondary traits. PheWAS results are then replicated using population-based UK Biobank data and compared across various



Funding: The presented research was funded by the National Cancer Institute at the National Institutes of Health (P30 CA046592 [LGF, LJB, MS, BM] and T32 CA83654 [RBP]) as well as by an Investigator Award of University of Michigan Precision Health (U063790 [LGF]). This material is based upon work supported by the National Science Foundation under Grant No. (NSF DMS 1712933). Any opinions, findings, and conclusions or recommendations expressed in this material are those of the author(s) and do not necessarily reflect the views of the National Science Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

PRS construction methods. We develop an accompanying visual catalog called *PRSweb* that provides detailed PheWAS results and allows users to directly compare different PRS construction methods.

Author summary

In the study of genetically complex diseases, polygenic risk scores (PRS) synthesize information from multiple genetic risk factors to provide insight into a patient's inherited risk of developing a disease based on his/her genetic profile. These risk scores can be explored in conjunction with health and disease information available in electronic medical records. PRS may be associated with diseases that may be related to or precursors of the underlying disease of interest. In this paper, we demonstrate how PRS can be used in concert with the medical phenome to better understand the etiology of disease subtypes nested within a broad disease classification. This is done by examining the shared and distinct genetic risk factors across the related but heterogeneous disease subtypes and also through our comparison of the secondary associations across the phenome corresponding to the subtype specific PRS. We consider several PRS construction methods in our study. This framework of analysis is enabled by access to electronic health records and genetics data. Leveraging and harnessing the rich data resources of the Michigan Genomics Initiative, a biorepository effort at Michigan Medicine, and the large population-based UK Biobank study, we investigated the primary and secondary disease associations with PRS constructed for the three most common types of skin cancer: melanoma, basal cell carcinoma and cutaneous squamous cell carcinoma.

Introduction

The underlying risk factors of genetically complex diseases are numerous. Genome-wide association studies (GWAS) on thousands of diseases and traits have made great strides in uncovering a vast array of genetic variants that contribute to genetic predispositions to disease [1]. In order to harness the information from a large number of genetic variants, a popular approach is to summarize their contribution through polygenic risk scores (PRS). While the performance of PRS to predict disease outcomes at a population level has been modest for many diseases, including most cancers, PRS have successfully been applied for risk stratification of cohorts [2, 3] and recently have been used to screen a multitude of clinical phenotypes (collectively called the medical phenome) for secondary trait associations [4, 5]. The goal of these phenome-wide screenings is to uncover phenotypes that share genetic components with the primary trait that, if pre-symptomatic, could shed biological insights into the disease pathway and inform early interventions or screening efforts for individuals at risk. Phenome-wide studies using PRS rely on an easily-calculated single biomarker that combines information across a spectrum of genetic variants. Additionally, PRS may be routinely available in patients' electronic health records (EHR) in the near future, making analyses based on PRS a useful route for agnostic interrogation of the medical phenome. Existing literature has explored how to construct PRS with respect to a single disease phenotype [6, 7]. In this paper, we demonstrate how polygenic risk scores can be used in concert with the medical phenome to better understand the etiology of disease subtypes nested within a broad disease classification. This is done by examining the shared and distinct genetic risk factors across the related but

heterogeneous disease subtypes and also through our comparison of the secondary associations across the phenome corresponding to subtype specific PRS.

In the post-GWAS era and with the availability of large biobank data from multiple sources, there is great interest in using gene-based biomarkers such as PRS for risk stratification and exploration of disease etiology. However, it is not always clear how best to construct a PRS for a particular phenotype. A PRS of the general form $\sum_{i=1}^{K} \hat{\beta}_i G_i$ requires specification of three things: a list of markers $G_1, G_2, \dots G_K$, the depth of the list or the number of markers (K), and the choice of the weights β_i . These choices can be based on information extracted from the latest GWAS or GWAS meta-analysis (when available), the NHGRI-EBI GWAS catalog of published results [1] (when available), or summary data for GWAS corresponding to each phenotype, e.g., from efforts that comprehensively screened the UK Biobank (UKB) phenome [8, 9]. While various methods for constructing PRS have been widely studied for predicting the primary phenotype collected through population-based sampling [6, 10], it is unknown how the different PRS will be associated with subtypes and related phenotypes and the associations PRS can help unravel across the medical phenome. The comparative performance of different PRS construction methods may depend on the phenotype of interest. For example, diseases such as depression, which are believed to involve genetic contribution of a large number of genetic variants, might perform differently than diseases such as cancer, which may involve a smaller number of causal variants. We provide important empirical results comparing different PRS construction methods in terms of their associations with secondary and related phenotypes and in terms of the associations they identify across the phenome.

In this paper, we first explore strategies for constructing a PRS using markers and weights obtained from either the latest GWAS or the NHGRI-EBI GWAS catalog that have reached genome-wide significance. We compare the PRS in terms of their performance [11] for the three most common skin cancer subtypes in the USA: basal cell carcinoma (MIM: 614740) [12], cutaneous squamous cell carcinoma [13] and melanoma (MIM: 155601) [14]. We compare the two strategies using an independent biobank of genetic, demographic, and phenotype data collected by the Michigan Genomics Initiative (MGI), a longitudinal biorepository effort within Michigan Medicine (University of Michigan) [4, 15]. Based on these results, we choose a PRS construction strategy for each skin cancer subtype for further analysis.

For the chosen PRS corresponding to each skin cancer subtype, we perform a phenomewide association study (PheWAS) relating the PRS to the EHR-based phenome of MGI. We call such a study a PRS-PheWAS [4]. PRS-PheWAS results are then replicated using the population-based UK Biobank data. In order to identify secondary associations that are not driven by the primary phenotype, we perform an additional "exclusion" PRS-PheWAS for each skin cancer subtype in which we exclude subjects with any type of observed skin cancer [4]. These studies demonstrate similarities and differences in PheWAS results for PRS constructed for different disease subtypes, suggesting that PRS constructed for various disease subtypes can provide insight into shared and unique secondary associations. Our results further demonstrate the ability of such studies to reproduce known associations between secondary phenotypes and particular disease subtypes through use of PRS.

We then describe an approach for using PRS to more directly evaluate the shared and unique genetic architecture of disease subtypes and identify shared and unique secondary phenotype associations related to this genetic architecture. We define a new PRS for each skin cancer subtype using loci *unique* to that subtype's chosen PRS. We further construct a composite PRS for general skin cancer consisting of loci *common* among all subtypes' PRS. While merging distinct clinical entities into a compound PRS may seem counterintuitive in terms of specificity, such an approach may increase power to identify dermatological features through PheWAS that are shared by all three subtypes. Such features may help provide insight into the shared biological etiology and disease development across disease subtypes.

The NHGRI-EBI GWAS catalog and latest GWAS PRS construction methods are based on published GWAS studies, which only report risk variants that reached genome-wide significance (usually defined by a P-value threshold of $P < 5x10^{-8}$). However, it is likely that there are additional risk variants below this threshold that could be associated with the trait but have not reached statistical significance [16]. Incorporating non-significant variants may conceivably improve the predictive power of a PRS but may also add additional random false positive signals, which in turn could dilute the discriminatory power of the true risk variants and diminish any predictive gain [6, 17]. To explore whether a PRS constructed using additional non-significant loci may perform differently than a PRS using only loci reaching genome-wide significance, we evaluated PRS constructed using publicly available genome-wide summary statistics from the UK Biobank at six different p-value thresholds both in terms of associations with skin cancer phenotypes and in terms of secondary phenotype associations. We further applied LDpred, a tool that adjusts GWAS summary statistics for the effects of linkage disequilibrium [7], to explore the performance of PRS incorporating the entire spectrum of available genetic information across the genome. There is extensive literature on constructing genomewide PRS using random effects, shrinkage methods, or thresholding (our focus) [7, 18, 19], but none of these methods have been evaluated in a PheWAS setting.

In this paper, we focus our attention on skin cancer, but the approaches used in this paper can be applied to other diseases with well-defined molecular subtypes (for example, breast cancer with ER and HER2-defined subtypes). We chose to use skin cancer as a demonstrative example for a variety of reasons. First, our discovery dataset (MGI) is particularly enriched for skin cancer cases due to the strong skin cancer clinical program at Michigan Medicine and due to the high rate of surgery for skin cancer patients. MGI primarily recruits participants undergoing surgery and is therefore enriched for cancers and other medical comorbidities when compared to a general population [4]. Additionally, skin cancer has well-defined subtypes, which allows us to explore performance of subtype-specific PRS. Skin cancer also provides a setting in which there may be genetic factors uniquely related to particular subtypes as well as genetic factors that are shared risk factors for all skin cancer subtypes. The various PRS-phenotype associations identified in this paper demonstrate ways to explore shared and subtype-specific phenotypes, and this joint framework may provide an enhanced understanding of the genome x phenome landscape.

We introduce an online visual web catalog called *PRSweb* that provides PRS-PheWAS results for melanoma, basal cell carcinoma, and squamous cell carcinoma. PheWAS results are available using four different PRS construction methods explored in this paper: latest GWAS, NHGRI-EBI GWAS catalog, the UK Biobank GWAS summary statistics using different significance thresholds, and LDpred. The weights and the marker list for each PRS method can be downloaded. Furthermore, PheWAS summary statistics can be accessed from *PRSweb* (see **Web resources**), providing future investigators with readily available and useful tools to perform further analyses.

Comprehensive phenome-wide and genome-wide analyses of large biobank studies with publicly available summary statistics can be rich resources for PRS construction, especially if the trait-of-interest's prevalence is high in the biobank. Using PRS, we can synthesize complex genetic information that can then be used to identify these shared genetic components across phenotypes. Compared to prior and existing literature, our contribution is new in four principal directions: (1) comparing various PRS construction methods in terms of their relationships with *related* EHR-derived phenotypes and subtypes (2) comparing PRS associations with secondary phenotypes across the phenome of MGI (academic medical center) and UK Biobank

(population-based), (3) developing PRS-based methods for understanding the shared and unique genetic contribution across disease subtypes both in terms of disease biology and in terms of secondary phenotype associations, and (4) introducing a publicly accessible online visual catalog PRSweb to visually represent the PRS x phenome landscape and access summary data from PheWAS.

Materials and methods

Discovery and replication cohorts

MGI cohort (discovery cohort). Participants were recruited through the Michigan Medicine health system while awaiting diagnostic or interventional procedures either during a preoperative visit prior to the procedure or on the day of the procedure that required anaesthesia. In addition to coded biosamples and secure protected health information, participants understood that all EHR, claims, and national data sources–linkable to the participant–may be incorporated into the MGI databank. Each participant donated a blood sample for genetic analysis, underwent baseline vital signs and a comprehensive history and physical assessment (also see Ethics Statement below). In the current study, we report results obtained from 30,702 unrelated, genotyped samples of recent European ancestry with available integrated EHR data (~90% of all MGI participants were inferred to be of recent European ancestry) [4].

UK Biobank cohort (replication cohort). The UK Biobank is a population-based cohort collected from multiple sites across the United Kingdom and includes over 500,000 participants aged between 40 and 69 years when recruited in 2006–2010 [20]. The open access UK Biobank data used in this study included genotypes, the Ninth and Tenth Revision of the International Statistical Classification of Diseases (ICD9 and ICD10) codes, inferred sex, inferred White British ancestry, kinship estimates down to third degree, birthyear, genotype array, and precomputed principal components of the genotypes.

Genotyping, sample quality control and imputation

MGI. DNA from 37,412 blood samples was genotyped on customized Illumina Infinium CoreExome-24 bead arrays and subjected to various quality control filters that resulted in a set of 392,323 polymorphic variants. Principal components and ancestry were estimated by projecting all genotyped samples into the space of the principal components of the Human Genome Diversity Project reference panel using PLINK (938 unrelated individuals) [21, 22]. Pairwise kinship was assessed with the software KING [23], and the software fastindep was used to reduce the data to a maximal subset that contained no pairs of individuals with 3rd-or closer degree relationship [24]. We also removed patients not of recent European descent from the analysis, resulting in a final sample of 30,702 unrelated subjects. Additional genotypes were obtained using the Haplotype Reference Consortium panel of the Michigan Imputation Server [25] and included over 17 million imputed variants with R² \geq 0.3 and minor allele frequency (MAF) \geq 0.1%. Genotyping, quality control and imputation are described in detail elsewhere [4]. Table 1 provides some descriptive statistics of the MGI and UK Biobank samples.

UK Biobank. The UK Biobank is a population-based cohort collected from multiple sites across the United Kingdom [20]. After quality control, we phased and imputed the 487,409 UK Biobank genotyped samples against the Trans-Omics for Precision Medicine (TOPMed) reference panel (see **Web resources**), which is composed of 60,039 multi-ethnic samples and 239,756,147 SNP and indel variants sequenced at high depth (30x). The phasing step was carried out on 81 chromosomal chunks with around 10,000 genotyped variants in each chunk using the software Eagle (with the "kbpwt" parameter set at 80,000) [26]. The imputation was carried out in 137 chromosomal chunks of around 20 Mbp in length with Mbp of total overlap

Characteristic	MGI	UK Biobank ^a			
n	30,702	408,961			
Females, n (%)	16,297 (53.1%)	221,052 (54.1%)			
Mean age, years (S.D.)	54.2 (15.9)	57.7 (8.1)			
Median number of visits per participant	27	not available			
Median days between first and last visit	1,469	not available			
Total number of ICD9 code days	3,459,331	49,085			
Number of unique ICD9 codes	10,323	3,126			
Median ICD9 code days per participant	58	2			
Total number of ICD10 code days	1,311,264	2,764,868			
Number of unique ICD10 codes	14,997	11,059			
Median ICD10 code days per participant	27	6			
Total number of PheWAS code days	6,367,117	3,679,624			
Number of unique PheWAS codes	1,856	1,680			
Median PheWAS code days per participant	94	8			
n samples without skin cancer diagnosis Mean age, years (S.D.) Females, n (%)	26,199 52.6 (15.8) 14,320 (54.7%)	395,179 57.7 (8.0) 214,237 (54.2%)			
n cases with skin cancer Mean age, years (S.D.) Females, n (%)	4,503 63.6 (13.3) 1,977 (43.9%)	13,782 (13,624 ^c) 62.0 (6.7) 6,815 (49.4%)			
n cases with melanomas of skin	1,772	2,724 (2,718 ^c)			
n cases with epithelial skin cancer and others ^b	3,220	11,152 (11,030 ^c)			
n cases with basal cell carcinoma	1,303	not available			
n cases with squamous cell carcinoma	836	not available			

Table 1. Demographics and	clinical characteristics	of the analytic datasets.
---------------------------	--------------------------	---------------------------

^a The provided characteristics are based a subset of White British subjects of the UK Biobank Study for which phenotype data and imputed data was available. To retain as many unrelated cases as possible for each trait, a maximal set of unrelated cases was identified before choosing controls from the pool of subjects unrelated to these cases or to each other.

^b Original PheWAS code "172.2" description "Other non-epithelial cancer of skin".

^c Unrelated cases

ICD9 and ICD10: International Statistical Classification of Diseases codes (9th and 10th revision), MGI based on code systemts with clinical modiciations ICD9-CM and ICD10-CM; S.D. standard deviation

https://doi.org/10.1371/journal.pgen.1008202.t001

on either side using the imputation tool Minimac4 (see **Web resources**). To increase computational efficiency, we imputed each of the chunks in batches of 10,000 samples at a time and then merged them back using BCFtools. Since Minimac4 imputes each sample independently, analyzing our samples in batches did not change their imputation estimates. However, this sampling would result in different imputation quality estimates for each batch, and thus we collapsed the estimates to generate imputation quality estimates across all the study samples. After imputation, we filtered out variants with estimated imputation accuracy of $R^2 < 0.1$, which left us with 177,895,992 variants.

Phenome generation

MGI. The MGI phenome was used as the discovery dataset and was based on ICD9 and ICD10 code data for 30,702 unrelated, genotyped individuals of recent European ancestry. These ICD9 and ICD10 codes were aggregated to form up to 1,857 PheWAS traits using the PheWAS R package (as described in detail elsewhere [4, 27]). For each trait, we identified case

and control samples. To minimize differences in age and sex distributions or extreme casecontrol ratios as well as to reduce computational burden, we matched up to 10 controls to each case using the R package "MatchIt" [28]. Nearest neighbor matching was applied for age and PC1-4 (using Mahalanobis-metric matching; matching window caliper/width of 0.25 standard deviations) and exact matching was applied for sex and genotyping array. A total of 1,578 casecontrol studies with >50 cases were used for our analyses of the MGI phenome.

UK Biobank. The UK Biobank phenome was used as a replication dataset and was based on ICD9 and ICD10 code data of 408,961 White British [15], genotyped individuals that were aggregated to PheWAS traits in a similar fashion (as described elsewhere [9]). To remove related individuals and to retain larger sample sizes, we first selected a maximal set of unrelated cases for each phenotype (defined as no pairwise relationship of 3^{rd} degree or closer [24, 29]) before selecting a maximal set of unrelated controls unrelated to these cases. Similar to MGI, we matched up to 10 controls to each case using the R package "MatchIt" [28]. Nearest neighbor matching was applied for birthyear and PC1-4 (using Mahalanobis-metric matching; matching window caliper/width of 0.25 standard deviations) and exact matching was applied for sex and genotyping array. A total of 1,366 case-control studies with >50 cases each were used for our analyses of the UK Biobank phenome.

On average, we were able to match 9.3 controls per case in the MGI phenome and 9.9 controls per case in the UKB phenome. Additional phenotype information for MGI and UK Biobank is included in **Fig B in** <u>S1 Text</u> and **Tables F-H in** <u>S1 File</u>.

Risk SNP selection

For each skin cancer subtype (melanoma, basal cell carcinoma, and squamous cell carcinoma), we generated four different sets of PRS: (1) based on merged summary statistics published in the NHGRI EBI GWAS catalog [1], (2) based on the latest available GWAS meta-analysis [30–32], (3) based on linkage disequilibrium (LD) clumping and p-value thresholding on publicly available GWAS summary statistics from the UK Biobank data [9], and (4) based on reweighting effect sizes of GWAS summary statistics by modeling LD with LDpred [7].

GWAS catalog SNP selection. We downloaded previously reported GWAS variants from the NHGRI-EBI GWAS catalog (file date: February 28, 2018) [1, 33]. None of the currently available skin cancer discovery studies included in the catalog used any subset of the MGI cohort or data from the UK Biobank. Single nucleotide polymorphism (SNP) positions were converted to GRCh37 using variant IDs from dbSNP: build 150 (UCSC Genome Browser) after updating outdated dbSNP IDs to their merged dbSNP IDs. Entries with missing risk alleles, risk allele frequencies, or odds ratios were excluded. If a reported risk allele did not match any of the reported forward strand alleles of a non-ambiguous SNP (not A/T or C/G) in the imputed genotype data (which correspond to the alleles of the imputation reference panel), we assumed minus strand designation and corrected the effect allele to its complementary base of the forward strand. Entries with a reported risk allele that did not match any of the alleles of an ambiguous SNP (A/T and C/G) in our data were excluded at this step. We only included entries with broad European ancestry (as reported by the NHGRI-EBI GWAS catalog). As a quality control check, we compared the reported risk allele frequencies (RAF) in controls with the RAF of 14,770 MGI individuals who had no cancer diagnosis (for chromosome X variants, we calculated RAF in females only). We then excluded entries whose RAF deviated more than 15%. This chosen threshold is subjective and was based on clear differentiation between correct and likely flipped alleles on the two diagonals (see Fig A in S1 Text) as noted frequently in GWAS meta-analyses quality control procedures [34]. No p-value threshold was applied to accommodate GWAS entries that used discovery/replication-based approaches to define

statistical significance. For each analyzed cancer type, we extracted risk variants that were also present in our genotype data and estimated pairwise LD (correlation r^2) using the allele dosages of the corresponding controls. For pairwise correlated SNPs ($r^2>0.1$) or SNPs with multiple entries, we kept the SNP with the most recent publication date (and smaller *P* value, if necessary) and excluded the other (**Table I in** <u>S1 File</u>).

Selection of risk SNPs from latest GWAS. In a similar fashion, we extracted and filtered reported association signals from large GWAS meta-analyses on basal cell carcinoma [31], cutaneous squamous cell carcinoma [30] and melanoma [32] and further restricted our attention to GWAS associations with p-values $< 5x10^{-8}$ (Table I in S1 File).

Genome-wide SNP selection of UK-Biobank-based GWAS. We obtained GWAS summary statistics for the ICD9- and ICD10-based PheWAS codes "172" (skin cancer; 13,752 cases versus 395,071 controls), "172.11" (melanoma; 2,691 cases versus 395,071 controls), and "172.2" (non-epithelial skin cancer; 11,149 cases versus 395,071 controls) from a public download [9] (see Web resources). These GWAS analyzed up to 408,961 White British European-ancestry samples with generalized mixed model association tests that used the saddlepoint approximation to calibrate the distribution of score test statistics and thus could control for unbalanced case-control ratios and sample relatedness [9]. For each trait, we reduced these summary statistics to SNPs that were reported with minor allele frequencies > 0.5% and were also available for the MGI data. Next, we performed LD clumping of all variants with p-values $< 5x10^{-4}$ using the imputed allele dosages to obtain independent risk SNPs (LD threshold of $r^2 > 0.1$ and a maximal SNP distance of 1 Mb). We limited the LD calculations to 10,000 randomly selected, unrelated, White British individuals to reduce the computational burden. Finally, we created subsets of these independent SNPs with p-values $<5x10^{-9}$, $<5x10^{-8}$, $<5x10^{-7}$, $<5x10^{-6}$, $<5x10^{-5}$, and $<5x10^{-4}$ (Table J in S1 File).

As an alternative to the above LD clumping and p-value-thresholding approach of genomewide PRS, we used the software package LDpred [7] to reweight the effect size of each variant of GWAS summary statistics by using a prior on effect sizes and LD information from a reference panel. We randomly selected 5,000 unrelated, White British samples of the imputed UK Biobank genotype data as the LD reference panel. We used the summary statistics of the UK Biobank-based GWAS on skin cancer (PheWAS code "172") and melanoma (PheWAS code "172.11) and applied LDpred's default filters (SNPs only; overlap between summary statistics, LD reference panel and target panel; MAF > 1%; non-ambiguous allele combinations), which resulted in 6.4 million variants for each GWAS. For each of the GWAS summary statistics, we ran LDpred with an LD radius of 2,800 SNPs, which corresponds to an average 1 Mb window, and modelled six proportions of causal SNPs in the prior (100%, 10%, 1%, 0.1%, 0.01%, and 0.001%) to obtain six genome-wide SNP sets with LDpred-reweighted effect sizes.

Construction of the polygenic risk scores

For each of the obtained SNP sets for each trait, we constructed a PRS as the sum of the allele dosages of risk increasing alleles of the SNPs weighted by their reported or reweighted log odds ratios. Restated, the PRS for subject j in MGI was of the form $PRS_j = \sum_i \beta_i G_{ij}$ where *i* indexes the included loci for that trait, β_i is the log odds ratios retrieved from the external GWAS summary statistics for locus *i*, and G_{ij} is a continuous version of the measured dosage data for the risk allele on locus *i* in subject *j*. The PRS variable was created for each MGI and UKB participant. For comparability of effect sizes corresponding to the continuous PRS across cancer traits and PRS construction methods, we transformed each PRS of the corresponding analytical data set to the standard Normal distribution using "ztransform" in the R package "GenABEL" [35].

Statistical analysis

In this study, we first constructed PRS for skin cancer subtypes using either the latest GWAS or the corresponding entries of the GWAS catalog. To compare the association between PRS and skin cancer phenotypes across different PRS construction methods, we fit the following model for each PRS and skin cancer phenotype:

logit(P(Phenotype is present|PRS, Age, Sex, Array, PC)) = $\beta_0 + \beta_{PRS}$ PRS + β_{Age} Age + β_{Sex} Sex + β_{Array} Array + β PC,

where the PCs were the first four principal components obtained from the principal component analysis of the genotyped GWAS markers and where "Array" represents the genotyping array. Our primary interest is β_{PRS} , while the other factors (Age, Sex and PC) were included to address potential residual confounding and do not provide interpretable estimates due to the preceding application of case-control matching. Firth's bias reduction method was used to resolve the problem of separation in logistic regression (Logistf in R package "EHR") [<u>36–38</u>], a common problem for binary or categorical outcome models when a certain part of the covariate space has only one observed value of the outcome, which often leads to very large parameter estimates and standard errors.

We then evaluated each PRS's (1) ability to discriminate between cases and controls by determining the area under the receiver-operator characteristics (ROC) curve (AUC) using R package "pROC" [39]; (2) calibration using Hosmer-Lemeshow Goodness of Fit test in the R package "ResourceSelection" [40, 41]; and (3) accuracy with the Brier Score in the R package "DescTools" [42]. These evaluations did not adjust for additional covariates. These metrics were estimated using roughly 2/3 of the matched data as a test set after fitting the above model on the remaining 1/3 of matched data, which we will refer to as the training data. We used these metrics and the logistic regression results to choose a PRS construction method to use for each skin cancer subtype moving forward. We compare these measures for various PRSphenotype relationships for each phenotype separately, so the comparative performance of these measures is not biased by the different case-control sampling. To explore the impact of incorporating non-significant loci into the PRS construction, we further performed the above analyses with PRS constructed using UK Biobank GWAS summary statistics with different pvalue thresholds. Similarly, we compared the LDpred-based PRS that assumed six different fractions of causal variants (non-zero effects) in the prior: 100%, 10%, 1%, 0.1%, 0.01%, and 0.001%. For LDpred comparisons we also report Nagelkerke's pseudo-R² to be consistent with the LDpred workflow [7].

Using the chosen PRS for each subtype, we conducted two PheWAS to identify other phenotypes associated with the PRS first for the 1,578 phenotypes in MGI and then for the 1,366 phenotypes from UK Biobank. To evaluate PRS-phenotype associations, we conducted Firth bias-corrected logistic regression by fitting a model of the above form for each phenotype and data source. Age represents the birth year in UK Biobank. To adjust for multiple testing, we applied the conservative phenome-wide Bonferroni correction according to the analyzed Phe-WAS codes ($n_{MGI} = 1,578$ or $n_{UK Biobank} = 1,366$). In Manhattan plots, we present–log10 (*p*value) corresponding to tests of H_0 : $\beta_{PRS} = 0$. Directional triangles on the PheWAS plot indicate whether a phenome-wide significant trait was positively (pointing up) or negatively (pointing down) associated with the PRS.

To investigate the possibility of the secondary trait associations with PRS being completely driven by the primary trait association, we performed a second set of PheWAS after excluding individuals affected with the primary or related cancer traits for which the PRS was constructed, referred to as "exclusion PRS PheWAS" as described previously [4]. We then

constructed new PRS scores representing shared and subsite-unique genetic components and performed a PheWAS for each.

To evaluate the impact of the matching in the PRS PheWAS and exclusion PRS PheWAS analyses in more concrete terms, we performed sensitivity analyses in which we conducted the PheWAS analyses using the unmatched data.

To evaluate how well prior presence of a secondary diagnosis can identify subjects with increased risk of developing skin cancer, we created a binary variable taking the value 1 if a given subject (1) was diagnosed with the secondary diagnosis and then diagnosed with skin cancer at least 365 days after or (2) was diagnosed with the secondary diagnosis and never diagnosed with skin cancer. We then fit a Firth bias-corrected logistic regression of the following form:

logit(P(Primary phenotype is present|Predictor, Age, Sex, Array, PC))

 $= \beta_0 + \beta_{PRS} I(\text{Secondary non skin cancer trait}) + \beta_{Age} Age + \beta_{Sex} \text{Sex} + \beta_{Array} \text{Array} + \boldsymbol{\beta} \text{ PC}$

where Array and PC were defined as before. Unless otherwise stated, analyses were performed using R 3.4.4 [43].

Development of an online visual catalog: PRSweb

The online open access visual catalog *PRSweb* available at <u>http://statgen.github.io/PRSweb</u> was implemented using "Pandas", a Data Analysis Library, which offers high level performance for large data structures and data analysis in the Python3 environment [44]. In combination with "Jinja2", a templating language for Python, and "Bootstrap", a Cascading Style Sheets (CSS) framework (see Web resources), static HTML files were compiled and allow easy and fast host-ing of all PRS-PheWAS results. The interactive plots are drawn with the JavaScript library "LocusZoom.js" (see Web resources) which offers dynamic plotting, automatic plot sizing and label positioning.

Ethics statement

Data were collected according to Declaration of Helsinki principles. MGI study participants' consent forms and protocols were reviewed and approved by the University of Michigan Medical School Institutional Review Board (IRB ID HUM00099605 and HUM00155849). Opt-in written informed consent was obtained.

Results

Assessing various PRS construction methods

We first explored the comparative performance of various PRS construction strategies in terms of the resulting PRS associations with related phenotypes in the skin cancer setting. <u>Table 2</u> provides the results.

PRS performance across related phenotypes. Using both the GWAS catalog and latest GWAS construction methods, the melanoma PRS was most strongly associated with the melanoma phenotype compared to the other phenotypes (based on the odds ratio). In particular, both melanoma PRS are more strongly related to the melanoma phenotype than the overall skin cancer phenotype, indicating that the reduced sample size associated with a more granular phenotype definition did not negatively impact the PRS specificity. Similarly, the two basal cell carcinoma PRS were most strongly associated with the basal cell carcinoma skin cancer subtype. Unlike the other cancer subtypes, the squamous cell carcinoma PRS did not appear to be

PRS (Number of SNPs)		Skin cancer n = 4,503 (1,501 / 3,002) ^d	Melanoma n = 1,896 (617 / 1,279) ^d	Basal cell carcinoma n = 1,303 (419 / 884) ^d	Squamous cell carcinoma n = 836 (273 / 563) ^d
PRS based on GWAS catalog				1	1
Melanoma (29)	$\begin{array}{c} PRS OR^{a} \\ P-value^{a} \\ AUC^{b} \\ HL \chi^{2}, P-value^{c} \\ Brier Score \end{array}$	$\begin{array}{c} 1.3 \left(1.26, 1.34\right)\\ 2.7 \mathrm{x10}^{53}\\ 0.57 \left(0.56, 0.58\right)\\ 6.8, 0.56\\ 0.14\end{array}$	1.48 (1.41,1.56) 1.3x10 ⁻⁵⁶ 0.61 (0.59,0.62) 7.6, 0.47 0.093	1.3 (1.23,1.38) 7.3x10 ⁻¹⁹ 0.58 (0.56,0.6) 12, 0.13 0.093	$\begin{array}{c} 1.23 \ (1.14,1.32) \\ 4.3x10^{-8} \\ 0.55 \ (0.53,0.58) \\ 2.2, \ 0.97 \\ 0.091 \end{array}$
Basal cell carcinoma (32)	$\begin{array}{c} PRS OR^{a} \\ P-value^{a} \\ AUC^{b} \\ HL \chi^{2}, P-value^{c} \\ Brier Score \end{array}$	$\begin{array}{c} 1.32 \ (1.27, 1.36) \\ 8x10^{-60} \\ 0.58 \ (0.56, 0.59) \\ 9.7, \ 0.29 \\ 0.14 \end{array}$	$\begin{array}{c} 1.31 \ (1.25, 1.37) \\ 7.2 \times 10^{-28} \\ 0.58 \ (0.56, 0.59) \\ 13, \ 0.1 \\ 0.093 \end{array}$	1.65 (1.56,1.75) 3.6x10 ⁻⁶⁵ 0.64 (0.62,0.66) 15, 0.066 0.091	$\begin{array}{c} 1.32 (1.23, 1.42) \\ 1.4x 10^{-14} \\ 0.58 (0.55, 0.6) \\ 12, 0.17 \\ 0.091 \end{array}$
Squamous cell carcinoma (18)	PRS OR ^a P-value ^a AUC ^b HL χ ² , P-value ^c Brier Score	$1.25 (1.21,1.3) \\ 4.8 \times 10^{-42} \\ 0.56 (0.55,0.57) \\ 8.1, 0.42 \\ 0.14$	$\begin{array}{c} 1.32 \ (1.26, 1.39) \\ 5.1 \times 10^{-31} \\ 0.58 \ (0.56, 0.59) \\ 6.2, \ 0.63 \\ 0.093 \end{array}$	$\begin{array}{c} 1.35 \ (1.28, 1.43) \\ 7.9 x 10^{-26} \\ 0.58 \ (0.56, 0.6) \\ 5.9, 0.66 \\ 0.093 \end{array}$	$1.26 (1.17,1.35) \\ 1.8x10^{-10} \\ 0.55 (0.53,0.58) \\ 18, 0.025 \\ 0.091$
PRS based on latest GWAS	·				·
Melanoma (20)	$\begin{array}{c} PRS OR^{a} \\ P-value^{a} \\ AUC^{b} \\ HL \chi^{2}, P-value^{c} \\ Brier Score \end{array}$	$\begin{array}{c} 1.31 \ (1.27, 1.36) \\ 3.5 \times 10^{.55} \\ 0.57 \ (0.56, 0.58) \\ 4.9, 0.76 \\ 0.14 \end{array}$	$\begin{array}{c} 1.49 \ (1.42, 1.57) \\ 1.2 x 10^{-55} \\ 0.6 \ (0.59, 0.62) \\ 13, 0.13 \\ 0.093 \end{array}$	1.39 (1.3,1.47) 7.9x10 ⁻²⁷ 0.59 (0.58,0.61) 16, 0.04 0.092	1.25 (1.16, 1.34) 4x10-9 0.55 (0.53, 0.58) 7.6, 0.47 0.091
Basal cell carcinoma (28)	PRS OR ^a P-value ^a AUC ^b HL χ^2 , P-value ^c Brier Score	$\begin{array}{c} 1.32 \ (1.28, 1.37) \\ 5.8 \times 10^{-61} \\ 0.58 \ (0.57, 0.59) \\ 5.9, 0.66 \\ 0.14 \end{array}$	$\begin{array}{c} 1.33 \ (1.27, 1.4) \\ 5.7 \times 10^{-32} \\ 0.58 \ (0.57, 0.6) \\ 20, 0.012 \\ 0.093 \end{array}$	$\begin{array}{c} 1.62 \ (1.53, 1.71) \\ 2.8 \times 10^{-60} \\ 0.63 \ (0.61, 0.65) \\ 6.5, 0.59 \\ 0.091 \end{array}$	$\begin{array}{c} 1.34(1.25,1.44)\\ 1.2x10^{-15}\\ 0.58(0.55,0.6)\\ 15,0.052\\ 0.091\end{array}$
Squamous cell carcinoma (10)	PRS OR ^a P-value ^a AUC ^b HL χ^2 , P-value ^c Brier Score	$1.34 (1.3, 1.38) \\ 1.1x10^{-70} \\ 0.58 (0.57, 0.6) \\ 13, 0.12 \\ 0.14$	$\begin{array}{c} 1.42 \ (1.36, 1.49) \\ 6.1 x 10^{-51} \\ 0.59 \ (0.58, 0.61) \\ 14, 0.082 \\ 0.093 \end{array}$	$\begin{array}{c} 1.47(1.4,1.56)\\ 1.8x10^{-43}\\ 0.61(0.59,0.63)\\ 7.1,0.53\\ 0.092\end{array}$	$\begin{array}{c} 1.4 \ (1.31, 1.5) \\ 2.1 x 10^{-21} \\ 0.59 \ (0.56, 0.61) \\ 6.4, \ 0.61 \\ 0.091 \end{array}$

Table 2. Associations of constructed PRS with skin cancer traits in MGI.

^a Association of each cancer with continuous PRS that were transformed to standard normal distribution. Point estimates, 95% confidence intervals and P- values are obtained by fitting Firth's bias-corrected logistic regression adjusted for age, sex, batch and PC1-4 to the full data.

^b Area under the curve of the receiver operating characteristic (ROC) curve with 95% confidence intervals calculated using the test data after fitting a model with the training data.

^c Hosmer-Lemeshow Goodness-of-Fit test for the test data after fitting a model with the training data

^d Number of cases in training / test set

https://doi.org/10.1371/journal.pgen.1008202.t002

most strongly associated with the squamous cell carcinoma phenotype. Instead, it was most strongly associated with basal cell carcinoma phenotype.

Comparisons across PRS construction methods. For each skin cancer phenotype, we compared the PRS-phenotype associations for various PRS. **Overall Skin Cancer**: PRS defined using different skin cancer subtypes had similar performance in terms of association with and discrimination for the overall skin cancer phenotype. By "discrimination," we refer to the ability of the PRS to distinguish cases and controls, which is measured by AUC. **Melanoma**: For the melanoma PRS, the GWAS catalog method and the latest GWAS method produced similar performance in terms of AUC, OR, Hosmer-Lemeshow goodness of fit, and Brier score. For example, the AUC for melanoma for the GWAS catalog melanoma PRS was 0.61 (95% CI, [0.59, 0.62]). The corresponding AUC for the latest GWAS method was 0.60 (95%

CI, [0.59, 0.62]). The two melanoma PRS out-performed PRS for the other skin cancer subtypes in terms of association with and discrimination for the melanoma phenotype. Fig C in S1 Text compares PRS weights to corresponding SNP-melanoma associations in MGI and UK Biobank. Basal Cell Carcinoma: As with melanoma, the basal cell carcinoma PRS produced similar results under the GWAS catalog and latest GWAS construction methods. The basal cell carcinoma AUC under both the GWAS catalog and latest GWAS methods were 0.64 (95% CI, [0.62, 0.66]) and 0.63 (95% CI, [0.61, 0.65]), respectively. The OR values and Brier score values were nearly identical, and neither approach produced evidence of lack of fit based on the Hosmer-Lemeshow statistic. Squamous Cell Carcinoma: The squamous cell carcinoma latest GWAS-based PRS was more strongly associated with the basal cell carcinoma phenotype than the squamous cell carcinoma GWAS catalog PRS, with an odds ratio of 1.4 (95% CI, [1.31, 1.5]). The squamous cell carcinoma phenotype using the GWAS catalog method produced a lower AUC (0.55, 95% CI [0.53, 0.58]) compared to the latest GWAS method (0.59, 95% CI [0.56, 0.61]). While a difference of 0.04 may not seem like a large difference in AUC in other applications, any improvement in AUC for PRS associations with observed phenotypes may be considered appreciable [45]. These two methods produced similar Brier scores.

Using the above comparisons between the various PRS, we chose a single PRS construction method for each skin cancer subtype to use in subsequent analyses. For melanoma and basal cell carcinoma, we chose the GWAS catalog method. While the GWAS catalog and latest GWAS methods were very similar for these two subtypes, we chose to pursue the GWAS catalog PRS for future analysis due to the larger number of loci for these PRS (29 vs 20 for melanoma and 32 vs 28 for basal cell carcinoma). We choose the latest GWAS method for squamous cell carcinoma due to its improved AUC and stronger OR compared to the GWAS catalog method. We will denote the chosen PRS for melanoma, basal cell carcinoma, and squamous cell carcinoma as mPRS, bPRS, and sPRS respectively.

PheWAS using the chosen PRS in MGI

Using each of the chosen PRS described above (mPRS, bPRS, and sPRS), we tested the association between each PRS and each of the 1,578 constructed phenotypes in MGI. For each PRS, the strongest associations were observed with dermatologic neoplasms that included overall skin cancer, melanoma, "other non-epithelial cancer of skin" (the PheWAS parent category of basal and squamous cell carcinoma), and carcinoma in situ of skin. In addition, secondary dermatologic traits such as actinic keratosis (with parent category "degenerative skin conditions and other dermatoses"), chronic dermatitis due to solar radiation (with parent category "dermatitis due to solar radiation"), and seborrheic keratosis were found to be associated with all three PRS (Fig 1 and Table K in S1 File). "Diseases of sebaceous glands", "sebaceous cyst", and "scar conditions and fibrosis of skin" were associated with bPRS. mPRS was most strongly associated with the melanoma phenotype (OR 1.48, 95% CI [1.41, 1.56]), while bPRS was most strongly associated with basal cell carcinoma (OR 1.65, 95% CI [1.56, 1.75]) followed closely by "other non-epithelial cancer of the skin" (OR 1.39, 95% CI [1.34, 1.44]). sPRS was most strongly associated with overall skin cancer (OR 1.34, 95% CI [1.3, 1.38]). The OR of all these phenotypes indicated an increased risk for primary and secondary traits with increasing PRS.

Validation of PRS-PheWAS in UK Biobank

To substantiate the detected dermatologic associations, we reiterated the association screen of the three PRS in the matched phenome of the population-based UK Biobank data set (Fig 1). In general, stronger evidence for association was found in UKB compared to MGI. This may





https://doi.org/10.1371/journal.pgen.1008202.g001

be driven by the larger sample sizes, e.g. a total of 13,623 skin cancer cases versus 4,503 in MGI. In the UK Biobank phenome, the large majority of the previous associations with dermatologic neoplasms were validated with the exception of the trait "dermatitis due to solar radiation", which had substantially fewer cases in UKB compared to MGI (390 versus 2,959 cases). Unlike MGI, all three PRS were significantly associated (at the phenome-wide level) with "cancer, suspected or other" and "malignant neoplasm, other." bPRS and sPRS were both associated with "diseases of the sebaceous glands" and "sebaceous cyst."

Exclusion PheWAS using the chosen PRS in MGI

In order to explore whether the identified PRS-phenotype associations were driven by the primary trait used to define the PRS (for example, as a side effect of treatment given after diagnosis with the primary trait), we performed a PheWAS for each PRS in which we excluded subjects who were cases for the primary trait or other skin cancer subtypes [4]. Results are shown in **Table K in** <u>S1 File</u> and **Fig D in** <u>S1 Text</u>. Actinic keratosis, a skin condition believed to be a precursor to non-melanoma skin cancers, remained significantly associated with the squamous cell carcinoma PRS in MGI and all three PRS in UK Biobank [46–48]. No other phenotypes were significant for MGI. "Sebaceous cyst" and its parent category "diseases of the sebaceous gland" were significant in the main UK Biobank PheWAS and remained significantly associated with basal cell carcinoma PRS and squamous cell carcinoma PRS in UK Biobank in the Exclusion PheWAS. **Appendix 1 in** <u>S1 Text</u> provides additional information on a sub-analysis of actinic keratosis as a predictor for future skin cancer.

PRS-PheWAS for shared and unique loci across skin cancer subtypes

In the PRS-PheWAS analyses, we note a striking overlap in the secondary dermatological traits significantly associated with each of the three PRS (mPRS, bPRS, sPRS). One potential explanation for this is that subjects may have more screening after an initial skin cancer diagnosis. Indeed, many subjects have multiple skin cancer diagnoses (**Fig F in** <u>S1 Text</u>). <u>Fig 2</u> shows the number of risk loci shared by different PRS. Six risk loci are shared between the mPRS, bPRS, and sPRS.

This observation inspired a follow-up exploration in which we defined a PRS for each cancer subtype using the loci unique to that subtype's chosen PRS. We call these new PRS scores mPRS-u, bPRS-u, and sPRS-u, which reflect the unique loci in the PRS for melanoma, basal cell carcinoma, and squamous cell carcinoma respectively. We also define a PRS consisting of all loci shared across the three skin cancer subtypes, which we call the shared PRS.

Table C in <u>S1 Text</u> shows the association between the various constructed PRS and the skin cancer phenotypes. As with mPRS, mPRS-u was most strongly associated with the melanoma phenotype. The bPRS-u score was similarly most strongly associated with basal cell carcinoma. We note that the melanoma AUC for the mPRS score was 0.61 (95% CI, [0.59, 0.62]) and is only 0.54 (95% CI, [0.52, 0.56]) for the mPRS-u score. Similarly, the basal cell carcinoma AUC for the bPRS score was 0.64 (95% CI, [0.62, 0.66]) and is only 0.57 (95% CI, [0.55, 0.59]) for the bPRS-u score. The sPRS-u score is not more strongly associated with the squamous cell carcinoma phenotype than the other skin cancer subtypes. For this reason, we do not include this PRS in further analyses. The shared PRS constructed as the unweighted sum of risk alleles of loci present in all three PRS scores (mPRS, bPRS, and sPRS) is more strongly associated with all three subtype phenotypes than the overall skin cancer phenotype.

Fig H in <u>S1 Text</u> shows PRS-PheWAS results using mPRS-u and bPRS-u. The scores again reveal their subtype specificity, while no notable secondary associations were observed. Although not shown here, additional exploration into the loci identified uniquely for each subtype, e.g. via pathway enrichment analyses, may provide some insight into subtype-specific biological mechanisms. **Fig I in** <u>S1 Text</u> shows PRS-PheWAS results for the shared PRS. Most strikingly, the shared skin cancer PRS was associated with the top skin cancer and dermato-logic traits that were previously found to be associated with the three partially overlapping PRS



Fig 2. Overlap between the three skin cancer trait loci. Reported risk SNPs within 1 Mb were merged into the same locus. Loci that were also reported to be associated with skin tanning ability are highlighted in bold. Loci were named according to the closest RefSeq genes (except *M1CR* a 385 kb locus with 16 RefSeq genes and *HV745896* named after a nearby, uncurated mRNA sequence).

https://doi.org/10.1371/journal.pgen.1008202.g002

constructs, suggesting that a shared genetic risk may be driving many of these secondary associations. These six underlying loci (*HERC2* [MIM 605837] /*OCA2* [MIM 611409], *IRF4* [MIM 601900], *MC1R* [MIM 155555], *RALY* [MIM 614663], *SLC45A2* [MIM 606202] and *TYR* [MIM 606933]) were previously found to be associated not only with skin cancer traits, but also with pigmentation traits of skin, eyes and hair (Fig 2; MIM 266300) [31, 49–68].

One of these pigmentation traits, skin tanning ability, the tendency of skin to sunburn rather than to suntan, is a well-known risk factor for all skin cancer traits [68, 69]. A PRS based on the independent risk variants of a recent GWAS meta-analysis on skin tanning ability [69] was strongly associated with overall skin cancer, melanoma, basal cell carcinoma, and squamous cell carcinoma and even outperformed the constructed PRS in some cases (**Table C** in <u>S1 Text</u>). Furthermore, the skin tanning ability PRS PheWAS identified a very similar set of traits as the shared skin cancer PRS, but in general displayed stronger associations (**Fig I in** <u>S1 Text</u>).

PRS construction based on UK Biobank summary statistics at different depths

To explore whether a PRS incorporating non-significant loci will outperform a PRS incorporating only significant loci, we constructed PRS using loci related to the phenotype at six different p-value thresholds based on publicly available GWAS summary statistics from the UK Biobank. Larger p-values indicate greater SNP depth (with more SNPs being incorporated into the PRS).

The ICD-code-based collection of UK Biobank GWAS results did not include basal cell carcinoma or squamous cell carcinoma subtypes; rather, it included only the merged trait 'nonepithelial cancer of skin' (**Fig B in** <u>S1 Text</u>). Thus, we limited our assessment of the summary statistics to the overall skin cancer GWAS and the melanoma GWAS (**Table J in** <u>S1 File</u>).

Table D in S1 Text provides the results. As with the other PRS construction methods, the melanoma PRS was most strongly associated with the melanoma phenotype for all p-value cutoffs except 5x10⁻⁴. For this p-value cutoff, the melanoma PRS had similar AUC and OR for the melanoma and basal cell carcinoma phenotypes. This p-value cutoff represents the least conservative inclusion cutoff with 1,193 included loci, and its results indicated that inclusion of too many suggestive SNPs at lower thresholds may reduce PRS performance. However, we also note that the most conservative cutoff $(5x10^{-9})$ produced a PRS that was based on only six loci, which had a weaker OR and AUC compared to other PRS created with less stringent cutoffs. The best performance in terms of AUC and OR relating to the melanoma phenotype were observed for p-value thresholds 5×10^{-7} and 5×10^{-8} , which included 13 and 9 loci respectively. The small number of loci identified by this method at more conservative p-value cutoffs may be driven by the lower sample size for melanoma in the UK Biobank compared to the published melanoma GWAS meta-analyses (n cases = 2,691 and n cases = 6,628 respectively). We note that the melanoma PRS constructed using the UK Biobank summary statistics produced lower AUC across all p-value thresholds than was seen for the latest GWAS and GWAS catalog PRS construction methods.

Among the skin cancer subtypes, the PRS constructed for overall skin cancer was most strongly associated with basal cell carcinoma across all p-value thresholds, with odds ratios ranging from 1.4 (95% CI [1.32, 1.48]) to 1.64 (95% CI [1.55, 1.74]). Among the PRS, the overall skin cancer PRS had the greatest discrimination for the overall skin cancer phenotype. Overall skin cancer and melanoma PRS had similar performance in terms of discrimination for the melanoma phenotype across various depths. The overall skin cancer PRS tended to be more strongly associated with and have similar or slightly better discrimination for the overall skin cancer PRS tended to the melanoma PRS, indicating that the overall skin cancer PRS was more accurate at predicting the overall skin cancer phenotype than the melanoma PRS. The overall skin cancer PRS had very similar association with and discrimination abilities for the overall skin cancer phenotype across all p-value thresholds except the least conservative ($p = 5x10^{-4}$), for which the AUC and odds ratio were smaller. Overall, the highest AUCs and strongest OR signals for both PRS and all skin cancer phenotypes were found at depths of $5x10^{-7}$ and $5x10^{-8}$.

In addition to associations with the primary and related phenotypes, we explored associations between PRS constructed at various UK Biobank summary statistic depths and secondary phenotypes. **Fig J** (overall skin cancer) and **Fig K** (melanoma) **in** <u>S1 Text</u> show PRS-PheWAS results in MGI using PRS constructed at different depths. Depths of $5x10^{-7}$ and $5x10^{-8}$ produced very similar results, and other depths tended to identify fewer phenotypes associated with the corresponding PRS. Phenotypes that were associated with the PRS at other depths had weaker associations. PRS-PheWAS using the melanoma PRS and the overall skin cancer PRS produced somewhat different results. For example, the melanoma PRS at different depths did not identify strong associations with "diseases of sebaceous glands", which is similar to previous PRS-PheWAS results for mPRS in MGI and UKB. In contrast, the overall skin cancer PRS did identify associations with "diseases of sebaceous glands" or its subcategories for all depths except $5x10^{-5}$ and $5x10^{-4}$. **Fig L in** <u>S1 Text</u> provides some additional information about the impact of depth on p-values for selected secondary associations.

PRS construction using LDpred

We evaluated the performance of PRS constructed using the LDpred algorithm, which incorporates millions of SNPs into the PRS definition. **Table E in** <u>S1 Text</u> provides results. For the overall skin cancer PRS, a modelled 1% proportion of causal variants produced the best results in terms of AUC and OR with respect to the overall skin cancer phenotype (OR 1.30, 95% CI [1.26, 1.35]) and AUC 0.58, 95% CI [0.56, 0.60]). For the melanoma PRS, a modelled 0.001% proportion of causal variants produced the best results with respect to melanoma (OR 1.42, 95% CI [1.36, 1.49]) and AUC 0.60, 95% CI [0.58, 0.62]). This LDpred melanoma PRS performed slightly better compared to the melanoma PRS constructed using UK Biobank summary statistics at a 5×10^{-8} depth in terms of associations with the primary phenotype. Using the PRS with a percentage of assumed causal variants producing the best pseudo R² statistic from **Table E in** <u>S1 Text</u>, we performed a PRS-PheWAS as shown in **Figure M in** <u>S1 Text</u>.

Table 3 summarizes the secondary phenotypes significantly associated with various PRS at the phenome-wide level. Many general skin cancer phenotypes are strongly associated with nearly all PRS. In particular, actinic keratosis and dermatitis due to solar radiation are associated with PRS for all three disease. In contrast, sebaceous cysts and "diseases of sebaceous glands" are strongly associated with PRS for basal cell carcinoma and squamous cell carcinoma but not with PRS for melanoma.

Online visual catalog: PRSweb

For comparison of the aforementioned PRS-PheWAS results and to provide researchers with resources for future PRS-based analyses, we developed an open access, online visual catalog *PRSweb* available at <u>https://statgen.github.io/PRSweb</u> that enables interactive exploration of the PheWAS results for each of the skin cancer subtypes and different PRS construction methods explored in this paper, for both the MGI and UK Biobank phenomes. *PRSweb* shows PRS-PheWAS plots with various choices of PRS in the drop-down menu (example screenshot in Fig 3) and offers downloadable PRS constructs (list of independent risk variants with corresponding weights). Mouse-over boxes offer detailed information about top results if needed, without impeding the overall user experience (grey box in Fig 3). Enrichment of cases in the upper quartiles of the PRS distribution are presented in forest plots.

Discussion

Polygenic risk scores combine information from a large number of genetic variants to stratify subjects in terms of their risk of developing a particular disease. In the first aim of this paper, we demonstrate that PRS can also be used to explore shared and unique genetic risk profiles and secondary phenotype associations for related disease subtypes. We focus our attention on the setting of skin cancer, but the statistical approach can be applied to study other diseases with well-defined molecular subtypes.

For each skin cancer subtype, we constructed PRS using various PRS construction methods and evaluated their associations to the overall skin cancer phenotype and the three most common skin cancer subtypes: melanoma, basal cell carcinoma, and squamous cell carcinoma. We demonstrated that PRS constructed using EHR-derived phenotypes can sometimes (but not always) have good performance in terms of specificity for the primary phenotype. All PRS were positively associated with all skin cancer phenotypes under consideration. This suggests that each individual PRS may capture some shared genetic contributions for disease development across skin cancer subtypes.

For each skin cancer subtype, we performed a PRS-PheWAS to identify secondary phenotypes that are associated with the corresponding PRS. We identified many dermatological

Table 3. Phenome-wide significant phenotypes in MGI identified using various PRS construction strategies^a.

		Melanoma								Basal cell carcinoma		Squamous cell carcinoma		
Method			PRS Constructed Using UK Biobank Summary LDpred Statistics ^b						mPRS	mPRS	bPRS	bPRS	sPRS	sPRS
Data Source				М	GI			MGI	MGI	UKB	MGI	UKB	MGI	UKB
Phenotype	PheCode	5x10 ⁻⁹	5x10 ⁻⁸	5x10 ⁻⁷	5x10 ⁻⁶	5x10 ⁻⁵	5x10 ⁻⁴	0.001% ^c						
Melanomas of skin	172.11	*	*	*	*	*	*	*	*	*	*	*	*	*
Melanomas of skin, dx or hx	172.1	*	*	*	*	*		*	*	*	*	*	*	*
Skin cancer	172	*	*	*	*	*		*	*	*	*	*	*	*
Other non-epithelial cancer of skin	172.2	*	*	*	*	*		*	*	*	*	*	*	*
Carcinoma in situ of skin	172.3	*	*	*	*	*		*	*	*	*	*	*	*
Basal cell carcinoma	172.21	*	*	*	*	*		*	*	n/a	*	n/a	*	n/a
Actinic keratosis	702.1	*	*	*	*			*	*	*	*	*	*	*
Benign neoplasm of skin	216	*	*	*	*			*	*	*			*	
Squamous cell carcinoma	172.22	*	*	*	*			*	*	n/a	*	n/a	*	n/a
Secondary malignant neoplasm of skin	198.7	*	*	*	*			*	*				*	
Disorder of skin and subcutaneous tissue NOS	689		*	*	*			*	*	*	*	*	*	*
Benign neoplasm of lymph nodes	229.1		*	*	*			*	*					
Neoplasm of uncertain behavior of skin	173		*	*	*			*	*		*		*	
Degenerative skin conditions & other dermatoses	702		*	*				*	*	*	*	*	*	*
Secondary malignancy of lymph nodes	198.1		*	*				*	*					
Dermatitis due to solar radiation	938		*	*				*	*		*		*	*
Chronic dermatitis due to solar radiation	938.2		*	*				*	*		*		*	*
Benign neoplasm of unspecified sites	229		*	*				*						*
Secondary malignant neoplasm	198		*	*				*	*					
Scar conditions and fibrosis of skin	701.2									*	*	*	*	*
Seborrheic keratosis	702.2								*	*	*	*	*	
Sebaceous cyst	706.2										*	*		*
Malignant neoplasm, other	195.1									*		*		*
Cancer, suspected or other	195									*		*		*
Other hypertrophic and atrophic conditions of skin	701									*		*		*
Diseases of sebaceous glands	706											*		*
Diseases of hair and hair follicles	704											*		*
Diaphragmatic hernia	550.2											*		*
	# SNPs:	6	9	13	27	156	1193	6.4x10 ⁶	29	29	32	32	18	18

* Indicates phenotype that reached phenome-wide significance in the corresponding PRS-PheWAS

^a Including only diseases identified in at least two PRS-PheWAS

^b Evaluated at different depths (p-value thresholds indicated below)

^c Modelled proportion of causal variants

Notes: mPRS, bPRS, and sPRS: chosen PRS for melanoma, basal cell carcinoma, and squamous cell carcinoma. mPRS and bPRS are based on GWAS catalog entries while sPRS is based on the single, latest GWAS results [30]; n/a: not available in ICD-based phenome of the UK Biobank.

https://doi.org/10.1371/journal.pgen.1008202.t003

features in addition to the primary phenotype, indicating the ability of PRS to reproduce associations with the primary phenotype even after multiple testing corrections and covariate adjustment. The majority of these associations were replicated in a PRS-PheWAS performed for the UK Biobank phenome. Our analyses identified actinic keratosis, which is believed to be



Fig 3. Example view from PRSweb (see web resources). A selection menu on top allows selection of PRS constructs and phenome while interactive plots with "PheWAS results" and "Exclusion PheWAS results" are generated after selection. "Associations between PRS and Selected Phenotype" plots are generated after clicking on a triangle in the PheWAS plots. Detailed summary statistics for each trait association are provided in mouseover elements (shown in grey). Underlying weights of a selected PRS can be downloaded via bottons below the plots (blue).

https://doi.org/10.1371/journal.pgen.1008202.g003

a precursor to squamous cell and basal cell carcinoma, as an independent predictor of basal cell and squamous cell carcinoma, and we demonstrated that incorporating the PRS in addition to clinical information improved discrimination for future skin cancer diagnoses (**Appendix 1 in** <u>S1 Text</u>) [46–48]. Additionally, some secondary phenotypes (for example, diseases of the sebaceous glands and sebaceous cysts) were identified in PRS-PheWAS only for the non-melanoma subtypes, which may provide some insight into some differentiating features of these subtypes.

In an additional analysis, we identified loci that were shared among all three skin cancer subtypes' PRS. Loci overlap between the PRS for the three subtypes may indicate factors related to common biology between the subtypes. We noted that all shared loci (*HERC2*/ *OCA2*, *IRF4*, *MC1R*, *RALY*, *SLC45A2* and *TYR*) were also loci that had been associated with human pigmentation traits and/or harbor key genes of the biochemical pathway of

melanogenesis [49, 53–61, 63, 66–70]. To more directly explore secondary associations common to all skin cancers, we constructed PRS using SNPs shared by all three skin cancer subtypes and a PRS for skin tanning ability using results from a recent GWAS meta-analysis.[69] The skin tanning ability PRS PheWAS identified a very similar set of traits to the shared PRS PheWAS, suggesting that the shared genetic component may in part represent genetic factors influencing the skin pigmentation and the skin reaction to sun exposure. In an attempt to more directly identify secondary associations unique to each subtype, we also constructed PRS using SNPs unique to each subtype's PRS. This analysis did not identify any strong subtypespecific associations, perhaps suggesting that the main genetic drivers of skin cancer are shared across subtypes.

In this paper, we explore strategies for constructing a PRS using markers and weights obtained from various publicly-available sources. We compare three general strategies for PRS construction. In the first strategy, we consider PRS constructed using a small number of markers and weights identified from either the latest GWAS or GWAS meta-analysis or the NHGRI-EBI GWAS catalog. We first compare these two PRS construction methods in terms of their associations with related and unrelated EHR-derived phenotypes. A priori, we may have some belief that the latest (and often the largest) GWAS may provide a better source of evidence to use for PRS construction due to larger sample sizes and (potentially) more carefully curated data. The latest GWAS and GWAS catalog methods produced PRS with similar performance in terms of their associations with and discrimination for the primary phenotype used to construct the PRS for both basal cell carcinoma and melanoma. The latest GWAS method produced better results for squamous cell carcinoma.

In the second PRS construction strategy, we use UK Biobank summary statistics at different p-value depths to construct PRS. We found that incorporating additional loci that did not reach genome-wide significance did improve the PRS' ability to discriminate cases from controls for the primary phenotype up to a point. In particular, PRS constructed using SNPs with p-values less than 5×10^{-8} or 5×10^{-7} resulted in the best performance, but further increasing the p-value threshold resulted in reduced performance. Crucially, we also observed stronger associations between the PRS and secondary phenotypes for PRS constructed using depths of 5×10^{-8} and 5×10^{-7} . These results suggest that some benefit may be observed by incorporating loci that do not reach significance into the PRS construction but incorporating too many loci with larger p-values may not improve the predictive ability of the PRS (for both primary and secondary phenotypes). However, this gain or reduction in PRS performance may depend on the phenotype of interest and on the prevalence of the phenotype in the analytical sample.

In the third PRS construction strategy, we use the LDpred method to construct PRS using the whole spectrum of observed genetic information. For melanoma, the LDpred PRS which modelled smaller fractions of causal variants were favored over the ones modelling larger fractions. While the PRS construction with LDpred performed similarly to the various depth approach in our particular study, its computational cost was substantially higher. However, recent work indicated that LDpred might outperform pruning and thresholding approaches when larger training data sets are available [71].

In our study, PRS associations with the primary phenotype were generally stronger using the latest GWAS and GWAS catalog-based PRS than for the PRS constructed using LDpred or using UK Biobank summary statistics at different depths. Since the underlying case numbers in the discovery GWAS for the former were substantially larger than the case numbers in the UK Biobank GWAS, no direct comparison between approaches was possible, and also because the required full summary statistics of the contributing and larger skin cancer case-control studies of the latest GWAS and GWAS catalog entries were not made available. Additionally, these simpler PRS construction strategies appeared to more clearly differentiate between related subtypes than the genome-wide PRS construction methods. All three PRS construction strategies produced many similar secondary associations from PRS-PheWAS, as shown in <u>Table 3</u>. Overall, these simpler PRS construction methods worked well in our particular skin cancer setting, and we did not see any improvements to using a much larger number of SNPs in the PRS construction. Future releases of full summary statistics from large skin cancer GWAS meta-analyses will enable more liberal thresholds and consequently may result in better performing PRS [3, 72, 73].

As a product of this study, we provide an online visual catalog *PRSweb* (see **Web resources**) that provides PRS-PheWAS results for the various skin cancer phenotypes for PRS constructed using the different methods explored in this paper. *PRSweb* will provide a routine way to compare different PRS construction methods and to explore PRS-PheWAS results in detail. Additionally, *PRSweb* provides the PRS construction details, which researchers can download and use in their own analyses. In the future, we plan to extend this online platform to include Phe-WAS for many other cancer phenotypes, which will make this online platform a general tool for identifying phenotypes related to particular types of cancer.

One limitation of the generalizability of this study comes from the homogeneous race profile of MGI and UK Biobank. UK Biobank consists of subjects of primarily European descent, and we restricted our analyses to subjects of European descent in MGI (excluding about 10% of the subjects in MGI) in order to ensure greater comparability between the two datasets. Additionally, many of the existing GWAS were conducted on European populations, and we wanted to consider similar samples when comparing the performance of PRS constructed using summary statistics from European populations. Unlike UK Biobank, MGI is not a population-based sample; rather, it is a sample of patients recruited from a large academic medical center. Patients were recruited prior to surgery through the anesthesiology department, and therefore they may present a potential for selection bias. PRS-PheWAS results are susceptible to collider bias caused by PRS relationships with both skin cancer diagnosis and other diseases related to sampling. The exclusion PheWAS strategy attempts to overcome this obstacle by evaluating PRS associations with secondary phenotypes only in the subjects without a skin cancer diagnosis. This approach does not remove the possibility of sampling bias, but it may help reduce the impact of sampling on the estimated PRS-phenotype associations. The chosen design of matched case-control studies can reduce bias and detection of false positives compared to the unmatched analysis, but it is typically less powerful than the unmatched analysis (Fig N in <u>S1 Text</u>).

An additional limitation for all EHR-based phenome-wide studies is the potential for bias due to phenotype misclassification. In **S1 Section 1**, we discuss this issue in more detail, and we provide a sensitivity analysis exploring the impact of misclassification on study results in **Fig O in** <u>S1 Text</u>. In spite of these limitations, a principled comparison of the various methods explored in this paper may provide researchers with a sense of the robustness of their PheWAS inference to the PRS construction method and an analytical framework for the exploration of shared genetic architecture of related traits.

Web resources

PRSweb; <u>https://statgen.github.io/PRSweb</u> University of Michigan Medical School Central Biorepository; <u>https://research.medicine.umich.edu/our-units/central-biorepository</u> UK Biobank; <u>http://www.ukbiobank.ac.uk/</u> UK Biobank GWAS summary statistics; <u>https://tinyurl.com/UKB-SAIGE</u> TOPMed variant browser, <u>https://bravo.sph.umich.edu/freeze5/hg38/</u>

TOPMed program, https://www.nhlbi.nih.gov/science/trans-omics-precision-medicinetopmed-program Minimac4; https://genome.sph.umich.edu/wiki/Minimac4 BCFtools; https://samtools.github.io/bcftools/bcftools.html KING; http://people.virginia.edu/~wc9c/ KING/ FASTINDEP; https://github.com/endrebak/fastindep PLINK; https://www.coggenomics.org/plink2/ Eagle; https://data.broadinstitute.org/alkesgroup/Eagle/ UCSC Genome Browser; http://genome.ucsc.edu/ R; https://cran.r-project.org/ NHGRI-EBI GWAS Catalog; https://www.ebi.ac.uk/gwas/ dbSNP; https://www.ncbi.nlm.nih.gov/projects/SNP/ Imputation server; https://imputationserver.sph.umich.edu/ Jinja, https://github.com/pallets/jinja Locuszoom, https://github.com/statgen/locuszoom.

Supporting information

S1 Text. Supporting material. This file contains supporting Appendices 1–2, Figures A-O and Tables A-E. (PDF)

S1 File. Supporting tables. This Excel file contains the following tables: Sheet 1: Table F, ICD9 codes to PheWAS code mapping tables for skin cancer; Sheet 2: Table G, ICD10 codes to PheWAS code mapping tables for skin cancer; Sheet 3: Table H, MGI and UK Biobank phenome summary; Sheet 4: Table I, GWAS Catalog and latest GWAS risk SNP selection; Sheet 5: Table J, UKB GWAS risk SNP selection; Sheet 6: Table K, Omnibus table of significant results.

(XLSX)

Acknowledgments

The authors acknowledge the University of Michigan Medical School Central Biorepository for providing biospecimen storage, management, and distribution services in support of the research reported in this publication. Part of this research has been conducted using the UK Biobank Resource under application number 24460.

Author Contributions

Conceptualization: Lars G. Fritsche, Lauren J. Beesley, Bhramar Mukherjee.

Data curation: Lars G. Fritsche, Peter VandeHaar, Robert B. Peng, Sarah A. Gagliano Taliun, Erin O. Kaleba, Thomas T. Klumpner, Stephanie E. Moser, Victoria M. Blanc.

Formal analysis: Lars G. Fritsche, Lauren J. Beesley, Peter VandeHaar, Robert B. Peng, Bhramar Mukherjee.

Funding acquisition: Lars G. Fritsche, Gonçalo R. Abecasis, Bhramar Mukherjee.

Investigation: Lars G. Fritsche, Lauren J. Beesley, Matthew Zawistowski, Sarah A. Gagliano Taliun, Stephen B. Gruber, Bhramar Mukherjee.

Methodology: Lars G. Fritsche, Lauren J. Beesley, Bhramar Mukherjee.

Project administration: Lars G. Fritsche, Lauren J. Beesley, Victoria M. Blanc, Chad M. Brummett, Sachin Kheterpal, Gonçalo R. Abecasis, Bhramar Mukherjee.

Resources: Lars G. Fritsche, Peter VandeHaar, Matthew Zawistowski, Sarah A. Gagliano Taliun, Sayantan Das, Jonathon LeFaive, Erin O. Kaleba, Thomas T. Klumpner, Stephanie E. Moser, Victoria M. Blanc, Chad M. Brummett, Sachin Kheterpal, Gonçalo R. Abecasis, Bhramar Mukherjee.

Software: Lars G. Fritsche, Peter VandeHaar, Robert B. Peng, Sayantan Das, Jonathon LeFaive, Bhramar Mukherjee.

Supervision: Lars G. Fritsche, Bhramar Mukherjee.

- Validation: Lars G. Fritsche, Lauren J. Beesley, Robert B. Peng.
- Visualization: Lars G. Fritsche, Lauren J. Beesley, Peter VandeHaar, Matthew Zawistowski, Bhramar Mukherjee.
- Writing original draft: Lars G. Fritsche, Lauren J. Beesley, Maxwell Salvatore, Sarah A. Gagliano Taliun, Stephen B. Gruber, Bhramar Mukherjee.
- Writing review & editing: Lars G. Fritsche, Lauren J. Beesley, Peter VandeHaar, Robert B. Peng, Maxwell Salvatore, Matthew Zawistowski, Sarah A. Gagliano Taliun, Sayantan Das, Jonathon LeFaive, Erin O. Kaleba, Thomas T. Klumpner, Stephanie E. Moser, Victoria M. Blanc, Chad M. Brummett, Sachin Kheterpal, Gonçalo R. Abecasis, Stephen B. Gruber, Bhramar Mukherjee.

References

- MacArthur J, Bowler E, Cerezo M, Gil L, Hall P, Hastings E, et al. The new NHGRI-EBI Catalog of published genome-wide association studies (GWAS Catalog). Nucleic Acids Res. 2017; 45(D1):D896– D901. Epub 2016/12/03. <u>https://doi.org/10.1093/nar/gkw1133</u> PMID: <u>27899670</u>.
- Wray NR, Goddard ME, Visscher PM. Prediction of individual genetic risk to disease from genome-wide association studies. Genome Res. 2007; 17(10):1520–8. Epub 2007/09/06. <u>https://doi.org/10.1101/gr. 6665407</u> PMID: <u>17785532</u>.
- Khera AV, Chaffin M, Aragam KG, Haas ME, Roselli C, Choi SH, et al. Genome-wide polygenic scores for common diseases identify individuals with risk equivalent to monogenic mutations. Nat Genet. 2018; 50(9):1219–24. Epub 2018/08/15. <u>https://doi.org/10.1038/s41588-018-0183-z</u> PMID: <u>30104762</u>.
- Fritsche LG, Gruber SB, Wu Z, Schmidt EM, Zawistowski M, Moser SE, et al. Association of Polygenic Risk Scores for Multiple Cancers in a Phenome-wide Study: Results from The Michigan Genomics Initiative. Am J Hum Genet. 2018; 102(6):1048–61. Epub 2018/05/22. <u>https://doi.org/10.1016/j.ajhg.2018</u>. 04.001 PMID: 29779563.
- Docherty AR, Moscati A, Dick D, Savage JE, Salvatore JE, Cooke M, et al. Polygenic prediction of the phenome, across ancestry, in emerging adulthood. Psychol Med. 2017:1–10. Epub 2017/11/28. <u>https://doi.org/10.1017/S0033291717003312</u> PMID: 29173193.
- Dudbridge F. Power and predictive accuracy of polygenic risk scores. PLoS Genet. 2013; 9(3): e1003348. Epub 2013/04/05. <u>https://doi.org/10.1371/journal.pgen.1003348</u> PMID: <u>23555274</u>.
- Vilhjalmsson BJ, Yang J, Finucane HK, Gusev A, Lindstrom S, Ripke S, et al. Modeling Linkage Disequilibrium Increases Accuracy of Polygenic Risk Scores. Am J Hum Genet. 2015; 97(4):576–92. Epub 2015/10/03. <u>https://doi.org/10.1016/j.ajhg.2015.09.001</u> PMID: <u>26430803</u>.
- Millard LAC, Davies NM, Gaunt TR, Davey Smith G, Tilling K. Software Application Profile: PHESANT: a tool for performing automated phenome scans in UK Biobank. Int J Epidemiol. 2017. Epub 2017/10/ 19. <u>https://doi.org/10.1093/ije/dyx204</u> PMID: <u>29040602</u>.
- Zhou W, Nielsen JB, Fritsche LG, Dey R, Gabrielsen ME, Wolford BN, et al. Efficiently controlling for case-control imbalance and sample relatedness in large-scale genetic association studies. Nat Genet. 2018; 50(9):1335–41. Epub 2018/08/15. <u>https://doi.org/10.1038/s41588-018-0184-y</u> PMID: <u>30104761</u>.
- Torkamani A, Wineinger NE, Topol EJ. The personal and clinical utility of polygenic risk scores. Nat Rev Genet. 2018. Epub 2018/05/24. <u>https://doi.org/10.1038/s41576-018-0018-x</u> PMID: 29789686.
- Steyerberg EW, Vickers AJ, Cook NR, Gerds T, Gonen M, Obuchowski N, et al. Assessing the performance of prediction models: a framework for traditional and novel measures. Epidemiology. 2010; 21 (1):128–38. Epub 2009/12/17. https://doi.org/10.1097/EDE.0b013e3181c30fb2 PMID: 20010215.
- Kauvar AN, Cronin T Jr, Roenigk R, Hruza G, Bennett R, American Society for Dermatologic S. Consensus for nonmelanoma skin cancer treatment: basal cell carcinoma, including a cost analysis of

treatment methods. Dermatol Surg. 2015; 41(5):550–71. Epub 2015/04/14. <u>https://doi.org/10.1097/</u>DSS.00000000000296 PMID: 25868035.

- Kallini JR, Hamed N, Khachemoune A. Squamous cell carcinoma of the skin: epidemiology, classification, management, and novel trends. Int J Dermatol. 2015; 54(2):130–40. Epub 2014/11/28. <u>https://doi.org/10.1111/ijd.12553</u> PMID: 25428226.
- Berwick M, Buller DB, Cust A, Gallagher R, Lee TK, Meyskens F, et al. Melanoma Epidemiology and Prevention. Cancer Treat Res. 2016; 167:17–49. Epub 2015/11/26. <u>https://doi.org/10.1007/978-3-319-22539-5_2</u> PMID: <u>26601858</u>.
- Bycroft C, Freeman C, Petkova D, Band G, Elliott LT, Sharp K, et al. Genome-wide genetic data on ~500,000 UK Biobank participants. bioRxiv. 2017. <u>https://doi.org/10.1101/166298</u>
- Reisberg S, Iljasenko T, Lall K, Fischer K, Vilo J. Comparing distributions of polygenic risk scores of type 2 diabetes and coronary heart disease within different populations. PLoS One. 2017; 12(7): e0179238. Epub 2017/07/06. <u>https://doi.org/10.1371/journal.pone.0179238</u> PMID: <u>28678847</u>.
- Evans DM, Visscher PM, Wray NR. Harnessing the information contained within genome-wide association studies to improve individual prediction of complex disease risk. Hum Mol Genet. 2009; 18 (18):3525–31. Epub 2009/06/26. <u>https://doi.org/10.1093/hmg/ddp295</u> PMID: <u>19553258</u>.
- Euesden J, Lewis CM, O'Reilly PF. PRSice: Polygenic Risk Score software. Bioinformatics. 2015; 31 (9):1466–8. Epub 2015/01/01. <u>https://doi.org/10.1093/bioinformatics/btu848</u> PMID: 25550326.
- So HC, Sham PC. Improving polygenic risk prediction from summary statistics by an empirical Bayes approach. Sci Rep. 2017; 7:41262. Epub 2017/02/02. <u>https://doi.org/10.1038/srep41262</u> PMID: 28145530.
- Sudlow C, Gallacher J, Allen N, Beral V, Burton P, Danesh J, et al. UK biobank: an open access resource for identifying the causes of a wide range of complex diseases of middle and old age. PLoS Med. 2015; 12(3):e1001779. Epub 2015/04/01. <u>https://doi.org/10.1371/journal.pmed.1001779</u> PMID: <u>25826379</u>.
- Wang C, Zhan X, Bragg-Gresham J, Kang HM, Stambolian D, Chew EY, et al. Ancestry estimation and control of population stratification for sequence-based association studies. Nat Genet. 2014; 46 (4):409–15. Epub 2014/03/19. https://doi.org/10.1038/ng.2924 PMID: 24633160.
- Li JZ, Absher DM, Tang H, Southwick AM, Casto AM, Ramachandran S, et al. Worldwide human relationships inferred from genome-wide patterns of variation. Science. 2008; 319(5866):1100–4. Epub 2008/02/23. https://doi.org/10.1126/science.1153717 PMID: 18292342.
- Manichaikul A, Mychaleckyj JC, Rich SS, Daly K, Sale M, Chen WM. Robust relationship inference in genome-wide association studies. Bioinformatics. 2010; 26(22):2867–73. Epub 2010/10/12. <u>https://doi.org/10.1093/bioinformatics/btg559</u> PMID: 20926424.
- Abraham KJ, Diaz C. Identifying large sets of unrelated individuals and unrelated markers. Source Code Biol Med. 2014; 9(1):6. Epub 2014/03/19. <u>https://doi.org/10.1186/1751-0473-9-6</u> PMID: 24635884.
- McCarthy S, Das S, Kretzschmar W, Delaneau O, Wood AR, Teumer A, et al. A reference panel of 64,976 haplotypes for genotype imputation. Nat Genet. 2016; 48(10):1279–83. Epub 2016/08/23. https://doi.org/10.1038/ng.3643 PMID: 27548312.
- Loh PR, Danecek P, Palamara PF, Fuchsberger C, YAR, HKF, et al. Reference-based phasing using the Haplotype Reference Consortium panel. Nat Genet. 2016; 48(11):1443–8. Epub 2016/10/28. <u>https://doi.org/10.1038/ng.3679</u> PMID: <u>27694958</u>.
- Carroll RJ, Bastarache L, Denny JC. R PheWAS: data analysis and plotting tools for phenome-wide association studies in the R environment. Bioinformatics. 2014; 30(16):2375–6. Epub 2014/04/16. https://doi.org/10.1093/bioinformatics/btu197 PMID: 24733291.
- Ho DE, Imai K, King G, Stuart EA. Matchlt: Nonparametric Preprocessing for Parametric Causal Inference. J Stat Softw. 2011; 42(8):1–28.
- Michailidou K, Lindstrom S, Dennis J, Beesley J, Hui S, Kar S, et al. Association analysis identifies 65 new breast cancer risk loci. Nature. 2017; 551(7678):92–4. Epub 2017/10/24. <u>https://doi.org/10.1038/ nature24284</u> PMID: 29059683.
- Chahal HS, Lin Y, Ransohoff KJ, Hinds DA, Wu W, Dai HJ, et al. Genome-wide association study identifies novel susceptibility loci for cutaneous squamous cell carcinoma. Nat Commun. 2016; 7:12048. Epub 2016/07/19. https://doi.org/10.1038/ncomms12048 PMID: 27424798.
- Chahal HS, Wu W, Ransohoff KJ, Yang L, Hedlin H, Desai M, et al. Genome-wide association study identifies 14 novel risk alleles associated with basal cell carcinoma. Nat Commun. 2016; 7:12510. Epub 2016/08/20. <u>https://doi.org/10.1038/ncomms12510</u> PMID: <u>27539887</u>.
- Ransohoff KJ, Wu W, Cho HG, Chahal HC, Lin Y, Dai HJ, et al. Two-stage genome-wide association study identifies a novel susceptibility locus associated with melanoma. Oncotarget. 2017; 8(11):17586– 92. Epub 2017/02/18. https://doi.org/10.18632/oncotarget.15230 PMID: 28212542.

- Welter D, MacArthur J, Morales J, Burdett T, Hall P, Junkins H, et al. The NHGRI GWAS Catalog, a curated resource of SNP-trait associations. Nucleic Acids Res. 2014; 42(Database issue):D1001–6. Epub 2013/12/10. <u>https://doi.org/10.1093/nar/gkt1229</u> PMID: 24316577.
- Winkler TW, Day FR, Croteau-Chonka DC, Wood AR, Locke AE, Magi R, et al. Quality control and conduct of genome-wide association meta-analyses. Nat Protoc. 2014; 9(5):1192–212. Epub 2014/04/26. <u>https://doi.org/10.1038/nprot.2014.071</u> PMID: 24762786.
- 35. GenABEL project developers. GenABEL: genome-wide SNP association analysis. 2013.
- 36. Heinze G. A comparative investigation of methods for logistic regression with separated or nearly separated data. Stat Med. 2006; 25(24):4216–26. Epub 2006/09/07. <u>https://doi.org/10.1002/sim.2687</u> PMID: <u>16955543</u>.
- 37. Heinze G, Ploner M, Dunkler D, Southworth H. logistf: Firth's bias reduced logistic regression. 2013.
- 38. Choi L, Beck C. EHR: Electronic Health Record (EHR) Data Processing and Analysis Tool. 2017.
- Robin X, Turck N, Hainard A, Tiberti N, Lisacek F, Sanchez JC, et al. pROC: an open-source package for R and S+ to analyze and compare ROC curves. BMC Bioinformatics. 2011; 12:77. Epub 2011/03/ 19. https://doi.org/10.1186/1471-2105-12-77 PMID: 21414208.
- 40. Hosmer DW, Lemeshow S. Applied Logistic Regression. New York, USA: John Wiley and Sons; 2010.
- Lele S, R., Keim JL, Solymos P. ResourceSelection: Resource Selection (Probability) Functions for Use-Availability Data. 2017.
- 42. Signorell A. DescTools: Tools for Descriptive Statistics. 2018.
- R Core Team. R: A Language and Environment for Statistical Computing. R Foundation for Statistical Computing, Vienna, Austria; 2016.
- 44. McKinney W. Data Structures for Statistical Computing in Python2010.
- **45.** Baker SG, Schuit E, Steyerberg EW, Pencina MJ, Vickers A, Moons KG, et al. How to interpret a small increase in AUC with an additional risk prediction marker: decision analysis comes through. Stat Med. 2014; 33(22):3946–59. Epub 2014/05/16. <u>https://doi.org/10.1002/sim.6195</u> PMID: <u>24825728</u>.
- 46. Fuchs A, Marmur E. The kinetics of skin cancer: progression of actinic keratosis to squamous cell carcinoma. Dermatol Surg. 2007; 33(9):1099–101. Epub 2007/09/01. <u>https://doi.org/10.1111/j.1524-4725.2007.33224.x</u> PMID: <u>17760601</u>.
- 47. Cohen JL. Actinic keratosis treatment as a key component of preventive strategies for nonmelanoma skin cancer. J Clin Aesthet Dermatol. 2010; 3(6):39–44. Epub 2010/08/21. PMID: 20725550.
- Jacobs RJ, Phillips G. Basal cell carcinoma mistaken for actinic keratosis. Clin Exp Optom. 2006; 89 (3):171–5. Epub 2006/04/28. <u>https://doi.org/10.1111/j.1444-0938.2006.00032.x</u> PMID: <u>16637973</u>.
- Adhikari K, Fontanil T, Cal S, Mendoza-Revilla J, Fuentes-Guajardo M, Chacon-Duque JC, et al. A genome-wide association scan in admixed Latin Americans identifies loci influencing facial and scalp hair features. Nat Commun. 2016; 7:10815. Epub 2016/03/02. <u>https://doi.org/10.1038/ncomms10815</u> PMID: <u>26926045</u>.
- Asgari MM, Wang W, Ioannidis NM, Itnyre J, Hoffmann T, Jorgenson E, et al. Identification of Susceptibility Loci for Cutaneous Squamous Cell Carcinoma. J Invest Dermatol. 2016; 136(5):930–7. Epub 2016/02/02. <u>https://doi.org/10.1016/j.jid.2016.01.013</u> PMID: <u>26829030</u>.
- Barrett JH, Iles MM, Harland M, Taylor JC, Aitken JF, Andresen PA, et al. Genome-wide association study identifies three new melanoma susceptibility loci. Nat Genet. 2011; 43(11):1108–13. Epub 2011/ 10/11. <u>https://doi.org/10.1038/ng.959</u> PMID: <u>21983787</u>.
- Bishop DT, Demenais F, Iles MM, Harland M, Taylor JC, Corda E, et al. Genome-wide association study identifies three loci associated with melanoma risk. Nat Genet. 2009; 41(8):920–5. Epub 2009/07/ 07. <u>https://doi.org/10.1038/ng.411</u> PMID: <u>19578364</u>.
- 53. Eriksson N, Macpherson JM, Tung JY, Hon LS, Naughton B, Saxonov S, et al. Web-based, participantdriven studies yield novel genetic associations for common traits. PLoS Genet. 2010; 6(6):e1000993. Epub 2010/06/30. <u>https://doi.org/10.1371/journal.pgen.1000993</u> PMID: 20585627 and own stock options in the company. 23andMe co-president AW has provided general guidance, including guidance related to the company's research undertakings and direction. PLoS Genetics' Editor-in-Chief Gregory S. Barsh is a potential consultant to 23andMe and therefore recused himself from the editorial and peer-review process. PLoS co-founder Michael B. Eisen is a member of the 23andMe Scientific Advisory Board.
- Han J, Kraft P, Nan H, Guo Q, Chen C, Qureshi A, et al. A genome-wide association study identifies novel alleles associated with hair color and skin pigmentation. PLoS Genet. 2008; 4(5):e1000074. Epub 2008/05/17. <u>https://doi.org/10.1371/journal.pgen.1000074</u> PMID: <u>18483556</u>.
- Hernandez-Pacheco N, Flores C, Alonso S, Eng C, Mak AC, Hunstman S, et al. Identification of a novel locus associated with skin colour in African-admixed populations. Sci Rep. 2017; 7:44548. Epub 2017/ 03/17. https://doi.org/10.1038/srep44548 PMID: 28300201.

- Jacobs LC, Hamer MA, Gunn DA, Deelen J, Lall JS, van Heemst D, et al. A Genome-Wide Association Study Identifies the Skin Color Genes IRF4, MC1R, ASIP, and BNC2 Influencing Facial Pigmented Spots. J Invest Dermatol. 2015; 135(7):1735–42. Epub 2015/02/24. <u>https://doi.org/10.1038/jid.2015.62</u> PMID: <u>25705849</u>.
- Law MH, Medland SE, Zhu G, Yazar S, Vinuela A, Wallace L, et al. Genome-Wide Association Shows that Pigmentation Genes Play a Role in Skin Aging. J Invest Dermatol. 2017; 137(9):1887–94. Epub 2017/05/16. <u>https://doi.org/10.1016/j.jid.2017.04.026</u> PMID: <u>28502801</u>.
- Lin BD, Mbarek H, Willemsen G, Dolan CV, Fedko IO, Abdellaoui A, et al. Heritability and Genome-Wide Association Studies for Hair Color in a Dutch Twin Family Based Sample. Genes (Basel). 2015; 6 (3):559–76. Epub 2015/07/18. <u>https://doi.org/10.3390/genes6030559</u> PMID: <u>26184321</u>.
- Liu F, Wollstein A, Hysi PG, Ankra-Badu GA, Spector TD, Park D, et al. Digital quantification of human eye color highlights genetic association of three new loci. PLoS Genet. 2010; 6(5):e1000934. Epub 2010/05/14. https://doi.org/10.1371/journal.pgen.1000934 PMID: 20463881.
- Liu F, Visser M, Duffy DL, Hysi PG, Jacobs LC, Lao O, et al. Genetics of skin color variation in Europeans: genome-wide association studies with functional follow-up. Hum Genet. 2015; 134(8):823–35. Epub 2015/05/13. https://doi.org/10.1007/s00439-015-1559-0 PMID: 25963972.
- Nan H, Kraft P, Qureshi AA, Guo Q, Chen C, Hankinson SE, et al. Genome-wide association study of tanning phenotype in a population of European ancestry. J Invest Dermatol. 2009; 129(9):2250–7. Epub 2009/04/03. <u>https://doi.org/10.1038/jid.2009.62</u> PMID: <u>19340012</u>.
- Nan H, Xu M, Kraft P, Qureshi AA, Chen C, Guo Q, et al. Genome-wide association study identifies novel alleles associated with risk of cutaneous basal cell carcinoma and squamous cell carcinoma. Hum Mol Genet. 2011; 20(18):3718–24. Epub 2011/06/28. <u>https://doi.org/10.1093/hmg/ddr287</u> PMID: 21700618.
- Rawofi L, Edwards M, Krithika S, Le P, Cha D, Yang Z, et al. Genome-wide association study of pigmentary traits (skin and iris color) in individuals of East Asian ancestry. PeerJ. 2017; 5:e3951. Epub 2017/11/08. <u>https://doi.org/10.7717/peerj.3951</u> PMID: <u>29109912</u>.
- Siiskonen SJ, Zhang M, Li WQ, Liang L, Kraft P, Nijsten T, et al. A Genome-Wide Association Study of Cutaneous Squamous Cell Carcinoma among European Descendants. Cancer Epidemiol Biomarkers Prev. 2016; 25(4):714–20. Epub 2016/02/26. <u>https://doi.org/10.1158/1055-9965.EPI-15-1070</u> PMID: 26908436.
- Song F, Amos CI, Lee JE, Lian CG, Fang S, Liu H, et al. Identification of a melanoma susceptibility locus and somatic mutation in TET2. Carcinogenesis. 2014; 35(9):2097–101. Epub 2014/07/02. <u>https:// doi.org/10.1093/carcin/bgu140</u> PMID: 24980573.
- Stokowski RP, Pant PV, Dadd T, Fereday A, Hinds DA, Jarman C, et al. A genomewide association study of skin pigmentation in a South Asian population. Am J Hum Genet. 2007; 81(6):1119–32. Epub 2007/11/14. <u>https://doi.org/10.1086/522235</u> PMID: <u>17999355</u>.
- Sulem P, Gudbjartsson DF, Stacey SN, Helgason A, Rafnar T, Magnusson KP, et al. Genetic determinants of hair, eye and skin pigmentation in Europeans. Nat Genet. 2007; 39(12):1443–52. Epub 2007/ 10/24. <u>https://doi.org/10.1038/ng.2007.13</u> PMID: <u>17952075</u>.
- Zhang M, Song F, Liang L, Nan H, Zhang J, Liu H, et al. Genome-wide association studies identify several new loci associated with pigmentation traits and skin cancer risk in European Americans. Hum Mol Genet. 2013; 22(14):2948–59. Epub 2013/04/04. https://doi.org/10.1093/hmg/ddt142 PMID: 23548203.
- Visconti A, Duffy DL, Liu F, Zhu G, Wu W, Chen Y, et al. Genome-wide association study in 176,678 Europeans reveals genetic loci for tanning response to sun exposure. Nat Commun. 2018; 9(1):1684. Epub 2018/05/10. https://doi.org/10.1038/s41467-018-04086-y PMID: 29739929.
- Sturm RA, Duffy DL. Human pigmentation genes under environmental selection. Genome Biol. 2012; 13(9):248. Epub 2012/11/01. <u>https://doi.org/10.1186/gb-2012-13-9-248</u> PMID: <u>23110848</u>.
- Márquez-Luna C, Gazal S, Loh P-R, Furlotte N, Auton A, Price AL. Modeling functional enrichment improves polygenic prediction accuracy in UK Biobank and 23andMe data sets. bioRxiv. 2018:375337. <u>https://doi.org/10.1101/375337</u>
- Chatterjee N, Wheeler B, Sampson J, Hartge P, Chanock SJ, Park JH. Projecting the performance of risk prediction based on polygenic analyses of genome-wide association studies. Nat Genet. 2013; 45 (4):400–5, 5e1–3. Epub 2013/03/05. <u>https://doi.org/10.1038/ng.2579</u> PMID: <u>23455638</u>.
- Mavaddat N, Michailidou K, Dennis J, Lush M, Fachal L, Lee A, et al. Polygenic Risk Scores for Prediction of Breast Cancer and Breast Cancer Subtypes. Am J Hum Genet. 2019; 104(1):21–34. Epub 2018/ 12/18. https://doi.org/10.1016/j.ajhg.2018.11.002 PMID: <u>30554720</u>.